\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}



\newif\ifnokeywords\nokeywordsfalse
\global\nokeywordstrue 
\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jrnlurlname{fjphs}
\usepackage{balance}

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}

\usepackage{qrcode}

%\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jshort{fjphs}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/index.php/\jrnlurlname}{\color{blue}\underline{\smash{https://pharmasprings.com/\jrnlurlname}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}


\makeatother


\def\citeauthor{~Jayachandra~Reddy~P}
\def\citetitle{Demonstration for the Presentation of a Clinical Case Report/Study}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}


\def\RunningAuthor{Jayachandra Reddy P }
\firstPage{77}
\articleType{Editorial Note}
\receivedDate{02 Mar 2021}
\acceptedDate{12 Mar 2021}
\revisedDate{08 Mar 2021}
\journalVolume{1}
\journalIssue{2}
\journalDoi{fjphs.v1i2.250}
\copyrightYear{2021}
\eISSN{2583-116X}
\journalDoi{fjphs.v1i2.250}
\def\authorCount{1}
\def\affCount{1}

\def\journalTitle{Future Journal of Pharmaceuticals and Health Sciences}

\title{Demonstration for the Presentation of a Clinical Case Report/Study}
\author{Jayachandra~Reddy~P\textsuperscript{*}~\\[5pt]{Associate Editor, Professor \& Principal, Krishna Teja Pharmacy College, Chadalawada Nagar, Renigunta Road, Tirupathi, Andhra Pradesh-517506, India}}

\begin{abstract}
The Abstract is the key part of the case study/ case report; it should be short and descriptive and addresses the main agenda of the report. It summarizes the all-over content to be effective and preferably less than 150 words. 
\end{abstract}\def\keywordstitle{Keywords}
\def\keywordstitle{}\begin{keywords}\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Jayachandra~Reddy~P~\\ Phone:\ 9032211999~\\  Email:\ pmanreddy@gmail.com
        \par}\vspace*{-13pt}\hrulefill\par\ifx\@eISSN\@empty\else{\fontsize{12}{14}\selectfont eISSN: \@eISSN\hfill pISSN: \@pISSN}\par\fi% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/10.26452/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont  Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
This section gave a rough overview of the topic focus. Background information should be collected from the relevant articles and note down the key points in the required format. Cite the references wherever needed. Review literature worth makes the case report to be effective \unskip~\cite{1028018:21599824,1028018:21599825}.

\textbf{Case study}

A case study can be defined as an intensive study about a patient with a detailed report of clinical manifestations, diagnosis, management, and follow-up \unskip~\cite{1028018:21599823}. Case reports defines usually a novel occurrence of a disease or rare conditions and some are focussing on ADR occurrence with a detailed study on how to manage them. Researchers describe how case studies examine complex phenomena in the natural setting to increase understanding of them.

\textbf{Case studies and case reports are written by? }

In the scientific and medical world, case studies and reports are written and submitted by doctors, nurses, pharmacists, medical researchers, postgraduate students of medicine, and healthcare-related students.

\textbf{What is the goal of a case study or a case report?}

The general purpose of a case study is to describe an individual situation (case) by a person, business, organization, or institute in detail. To identify the key issues of the case, to analyze the case using relevant theoretical concepts from the subject recommended a course of action for that particular case, particularly for problem-solving case studies.

\textbf{Case}

This part is the heart of a case report. This includes the details of the following parameters-


  
  \begin{enumerate}
  \item \relax Patient demographic details
  \item \relax History and chief complaints of the present case
  \item \relax Past medical and medication history
  \item \relax Physical and systemic examination results
  \item \relax Laboratory results
  \item \relax Treatment chart or plan
  \item \relax Treatment goals
  \item \relax Outcome results
  \end{enumerate}
  The author should maintain the privacy of patients while publishing the reports and excluded unnecessary details.

\textbf{Discussion}

This is the vital part of the study includes the extension of the introduction and case report on actually what the problem addresses. It should be crystal clear to make our study worthy of publication. This should be followed by discussing some other articles that support our theory and outcomes. This literature review makes us noteworthy and provides evidence to convince the journal for publication. It conveys our topic with proper proof to the readers. Citing the references is important in this section.
    
\section{Conclusion}
Our study finally ends with the conclusion part. It includes the key points of the topic and the author's perspectives on how to overcome them. It should short, straight forward and informative to the readers.

\textbf{Informed consent}

Patient informed consent is a valuable note and mandatory for the author to publish his/her article. It is a requirement for studies involving human subjects. If for case report on children, parental informed should be taken. For psychiatric cases, guardian or family member's informed consent was required to publish the study legally. Wherever pictures or photographs are needed to address the issue, be careful about the identity of the patient.
    

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
